Author's response to reviews

Title: To compare the efficacy of two kinds of Zhizhu pills in the treatment of functional dyspepsia of spleen-deficiency and qi-stagnation syndrome randomized group sequential comparative trial

Authors:

Hongli Wu (hlwcapri@yahoo.com.cn)
Zhiwei Jing (drizw@163.com)
Xudong Tang (txdly@sina.com)
Xinyue Wang (wxy6687@sina.com)
Shengsheng Zhang (zss2000@sohu.com)
Hongxin Cao (caohongxin58@yeah.net)
Yanan Yu (pumpkinnaicha@163.com)
Zhong Wang (zhonw@vip.sina.com)
Luqi Huang (huanglугi@263.net)
Youhua Yu (yuyh@mail.cacms.ac.cn)
Yongyan Wang (yongyanwang@126.com)

Version: 5 Date: 14 June 2011

Author's response to reviews: see over
Dear Editor,

Thank you so much for your E-mail letter dated June 13, 2011, regarding our manuscript entitled “To compare the efficacy of two kinds of Zhizhu pills in the treatment of functional dyspepsia of spleen-deficiency and qi-stagnation syndrome: a randomized group sequential comparative trial” (MS: 1992065822329852).

In the revised version we have addressed each of the concerns of the reviewer, our detailed response to the questions is enclosed, and the changes are highlighted in red in the revised manuscript. Now we believed that the manuscript has been improved and conforms to the journal style, and hope that it will be accepted for publication in BMC Gastroenterology. Finally, thank you for the help on our manuscript and look forward to hearing from you soon.

Sincerely yours,
Hongli Wu